Author's response to reviews

Title: Fluvastatin attenuates hepatic steatosis-induced fibrogenesis in rats through inhibiting paracrine effect of hepatocyte on hepatic stellate cells

Authors:

Lee-Won Chong (leewonch@ms52.hinet.net)
Yi-Chao Hsu (hsuyc@mmc.edu.tw)
Ting-Fang Lee (leetingfang@hotmail.com)
Yun Lin (u9223015@gmail.com)
Yung-Tsung Chiu (ytchiu@vghtc.gov.tw)
Kuo-Ching Yang (M001003@ms.skh.org.tw)
Jaw-Ching Wu (jcwu@vghtpe.gov.tw)
Yi-Tsau Huang (huangyt@nicm.edu.tw)

Version: 2
Date: 22 July 2014

Author's response to reviews:

July 22, 2014
Magdalena Morawska, PhD
Executive editor
BMC Gastroenterology

Dear editor,

We would like to submit the revised manuscript entitled “Fluvastatin attenuates hepatic steatosis-induced fibrogenesis in rats through inhibiting paracrine effect of hepatocyte on hepatic stellate cells” for your consideration for publication in BMC Gastroenterology.

In response to your request, we have added the competing interests and author contribution after the conclusion section. Line and page numbers have also been added.

Thank you very much.
Your kind consideration and editorial arrangement of our manuscript is deeply appreciated.

With my best regards,

Yours sincerely,

Yi-Tsau Huang, M.D., Ph.D.
Director, National Research Institute of Chinese Medicine,
Professor, Institute of Traditional Medicine, National Yang-Ming University,
No.155-1, Li-Nong Street, Sec. 2, Taipei 112, Taiwan.
Tel: +886-2-28225592
Fax: +886-2-28230488.
E-mail: huangyt@nricm.edu.tw